

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

So, Stella; Wong, Kar Shun; Zhang, Feida; Zhang, Xu

Article

Value relevance of proportionate consolidation versus the equity method: Evidence from Hong Kong

China Journal of Accounting Research

Provided in Cooperation with:

Sun Yat-sen University

Suggested Citation: So, Stella; Wong, Kar Shun; Zhang, Feida; Zhang, Xu (2018): Value relevance of proportionate consolidation versus the equity method: Evidence from Hong Kong, China Journal of Accounting Research, ISSN 1755-3091, Elsevier, Amsterdam, Vol. 11, Iss. 4, pp. 255-278, https://doi.org/10.1016/j.cjar.2018.06.002

This Version is available at: https://hdl.handle.net/10419/241780

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



https://creativecommons.org/licenses/by-nc-nd/4.0/





Contents lists available at ScienceDirect

China Journal of Accounting Research

journal homepage: www.elsevier.com/locate/cjar



Value relevance of proportionate consolidation versus the equity method: Evidence from Hong Kong



Stella So^a, Kar Shun Wong^a, Feida (Frank) Zhang^{b,*}, Xu Zhang^c

- a Department of Accountancy and Law, Hong Kong Baptist University, Hong Kong, China
- ^b Business School, University of Queensland, Australia
- ^c Department of Accounting and Information Management, University of Macau, Macau, China

ARTICLE INFO

Article history: Received 1 October 2016 Accepted 29 June 2018 Available online 3 September 2018

Keywords: Value relevance Proportional consolidation method Equity method Financial statements

ABSTRACT

Whether proportionate consolidation (PC) or the equity method (EM) provides more informative financial statements is a controversial issue. This study uses data from listed companies in Hong Kong to investigate the value relevance of the EM compared with PC during 2005–2008 when the local word-for-word equivalent HKAS 31 offered the same options. The results of this study provide evidence that PC does not offer higher value relevance than the EM. PC's horizontal aggregation of a portion of the operations, assets and liabilities of the jointly controlled entities with those of the venturer is less informative to investors than the EM's vertical aggregation.

© 2018 Sun Yat-sen University. Production and hosting by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

1. Introduction

This study investigates the value relevance of two alternative accounting methods, proportionate consolidation (PC) and the equity method (EM), which were introduced for reporting interests in jointly

E-mail addresses: stellaso@hkbu.edu.hk (S. So), wongks@hkbu.edu.hk (K.S. Wong), feida.zhang@uq.edu.au (Feida (Frank) Zhang), xuzhang@umac.mo (X. Zhang).

^{*} Corresponding author.

controlled entities¹ (JCEs) in International Accounting Standard No. 31 "Financial Reporting of Interests in Joint Ventures" (IAS 31 (1990)² (IASC, 1990; IASCF, 2010a). Consistent with the recent value relevance studies in the literature (e.g., O'Hanlon and Taylor, 2007; and as summarized in Barth et al., 2001), this study measures value relevance as the association between the key accounting amounts available from the companies' financial statements and their share prices.

In September 2007, as part of its efforts to eliminate choices in the existing standards (IASCF, 2010b), the International Accounting Standards Board (IASB) published Exposure Draft No. 9 "Joint Arrangements" (ED 9 (2007)) (IASCF, 2007a), which proposed to eliminate the PC option for reporting interests in JCEs from IAS 31 (1990), allowing only EM. Ironically, PC is the benchmark method in IAS 31 (1990) (IASC, 1990; IASCF, 2010a). An analysis of the comment letters received on ED 9 (2007) revealed, however, that about two thirds of the respondents disagreed with the proposed elimination of PC (Deloitte, 2008). The IASB was criticized for its haste in achieving convergence by adopting the U.S. standard (which allows only EM) without undertaking a detailed analysis of the accounting alternatives to select the most appropriate method (e.g., Knorr, 2008; Maes, 2008).

Joint venture arrangements provide ready access to new markets, expertise and technology, and the ability to reduce costs and spread risk (Koh and Venkatrman, 1991). A joint venture is an important form of business arrangement, making it essential to establish clear guidance on its reporting.

This study aims to investigate whether EM and PC result in financial statements that provide different value relevance for investors in the capital market. The Conceptual Framework identifies investors as the primary users of financial reports and the reports are primarily intended to facilitate their value estimations (IFRSF, 2010). Empirical studies specifically comparing EM and PC are sparse and their broad focus is on selected accounting ratios and more specifically on their differential accounting return predictability (Graham et al., 2003) and bond riskiness relevance (Kothavala, 2003; Stoltzfus and Epps, 2005; Bauman, 2007). By examining the value relevance of the key accounting figures derived from EM versus PC financial statements, this study provides a stronger and more comprehensive empirical input for the debate on ED 9 (2007), and on the suitability of the IASB's removal of the PC option from IAS 31 (1990).

In this study, we examine the value relevance of EM versus PC in the context of public companies listed on the Hong Kong Stock Exchange (HKSE) since 2005, for two reasons. First, since their full convergence on 1 January 2005, the local Hong Kong Financial Reporting Standards (HKFRS) replicate the International Financial Reporting Standards (IFRS) word-for-word and paragraph-for-paragraph. The Hong Kong Accounting Standards No. 31 "Interests in Joint Ventures," issued in December 2004 (HKICPA, 2004) (HKAS 31 (2004)), carries the same alternative options for interests in JCEs, and the results of this study can therefore serve as a strong reference for IAS 31 (1990) in the IFRS. Second, HKSE listers are likely to

¹ IAS 31 (1990) defines a JCE as a joint venture arrangement that involves the establishment of a corporation, partnership or other entity in which each venturer has an interest. "Joint Arrangements" in ED 9 (2007), however, proposes replacing the term "JCE" with "joint venture" (IASCF, 2007a). ED 9 (2007) defines "joint venture" as a joint arrangement that is jointly controlled by venturers who do not have rights to individual assets or obligations for expenses of the venture but rather, each venturer is entitled to a share of the outcome (e.g., profit or loss) of the activities of the joint venture. "Joint venture" in ED 9 (2007) therefore has essentially the same meaning as "JCE" in IAS 31 (1990), where the venturers' rights to the assets and liabilities of the venture are indirectly attained through their interests in the JCE. Throughout this paper, we use "JCE" to avoid an attempt to anticipate the outcome of ED 9 (2007), which is expected to be finalized in the first quarter of 2011 (IASB, 2011). The term "joint venture" is also used in this paper, but it refers to joint ventures in general. However, we admit that the definitions of "joint venture" under IAS 31 and ED 9 are not exactly the same. Under IAS 31, the definition is more based on the legal form of the joint arrangement. In contrast, under ED 9 the definition is mainly driven by the economic substance of the joint arrangement. This could be a caveat for our research.

² The two alternative accounting methods for JCEs (EM and PC) examined in this paper were first introduced in IAS 31 in 1990 by the former International Accounting Standards Committee (IASC). Although IAS 31 (1990) was reformatted in 1994, slightly amended in 1998, 1999 and 2000, re-adopted by the IASB in 2001, re-titled "Interests in Joint Ventures" in 2003, and further amended in 2004 and 2008 as a result of amendments in other accounting standards, no change has been made to the two alternative methods first introduced in 1990. Throughout this paper, reference is made to IAS 31 (1990), although it has been replaced by the subsequent updated versions and conveniently contained in the yearly bound volumes of the International Financial Reporting Standards (IFRSs) as approved at the beginning of each year.

have significant investments in joint ventures, especially those in mainland China,³ as a result of the latter's entry into the World Trade Organization, the popularity of Chinese–foreign joint ventures (Nippa et al., 2007) and the reported vast investments coming from Hong Kong in particular (Jiang, 2006). With the availability of more observations with more significant⁴ investments in joint ventures, the test power of this study will be higher.

Using models adapted from Easton and Harris (1991), Amir et al. (1993) and Barth (1994), this study provides evidence that PC does not offer higher value relevance than EM. Specifically, PC's horizontal aggregation of a portion of the operations, assets and liabilities of the JCEs is less informative to investors than EM's vertical aggregation.

This paper proceeds as follows. Section 2 describes joint venture accounting. Section 3 describes related studies and Section 4 details the research method. Section 5 discusses the empirical results, and the paper concludes in Section 6 with a summary and discussion of the research limitations.

2. Joint venture accounting

A JCE is an incorporated joint venture arrangement. Unlike investments in subsidiaries, control over a JCE is joint rather than exclusive and is shared with the other venturers through unanimous decisions on the financial and operating policies of the JCE (IASCF, 2010a, 2007a). It is the shared control over the essential activities of such an entity that gives rise to unique joint venture accounting issues.

2.1. Vertical versus horizontal aggregation

Technically, EM and PC are two very different methods and can significantly influence the amounts reported on the face of financial statements. The requirements for disclosure in the footnotes of the accounts are, however, essentially the same under the two methods.

IAS 31 (1990) allows a choice between the two methods but recommends PC as the benchmark method and permits EM as an alternative treatment. In contrast, the U.S. standard mandates EM (APB, 1971).

EM has been referred to as a "one-line consolidation," whereby the venturer's interests in the JCE's individual assets and liabilities and individual income and expense items are "vertically" aggregated into a single line item on the balance sheet (labeled as "interests in JCE," representing the venturer's share of the JCE's net assets) and income statement (labeled as "share of profit of JCE"), respectively. EM is illustrated in Panel A of Exhibit 1. Because the IFRS mandates EM for investments in associates, full details about EM are provided in IAS 28 "Investments in Associates" (IASCF, 2010c).

PC, in contrast, involves a "horizontal" line-by-line "bringing together" of the venturer's share of each of the individual assets and liabilities, income and expense items of the JCE with the venturer's own items (IASCF, 2010a). In taking a line-by-line approach, the method is similar to the full consolidation method as applied to subsidiaries, but dissimilar because only the venturer's share of the JCE is included.

In PC (IASCF, 2010a), two reporting formats may be used for the "bringing together." The venturer may aggregate the venturer's share of each of the individual assets and liabilities, income and expense items of the JCE horizontally with the venturer's own items line-by-line. Alternatively, the venturer may maintain them in separate line items. The two reporting formats are illustrated in Panels B1 and B2 of Exhibit 1 and are labeled as "combined" and "separate" formats, respectively.

³ Mainland China, Continental China and the Chinese mainland are geopolitical terms that refer to the area under the jurisdiction of the People's Republic of China, excluding Hong Kong and Macau, which run on different economic and political systems. This is the so-called one country, two systems.

⁴ As of 31 December 2008, 202 HKSE listers had interests in JCEs (comprising 23% of the total 867 HKSE listers with December year ends). Of the 342 observations (firm-years) in this study, 304 had JCEs in mainland China, of which 18 had 20–43 such JCEs and the mean number of mainland China JCEs per observation was 4.44.

⁵ An associate is an entity over which the investor has significant influence and is neither a subsidiary nor an interest in a joint venture (IASCF, 2010c). It is presumed that an investor has significant influence if the investor holds, directly or indirectly (e.g. through subsidiaries), 20% or more of the voting power of the investee, unless it can be clearly demonstrated that this is not the case (IASCF, 2010c).

The separate format of PC is, however, rarely used in practice, presumably because it results in large and cumbersome statements. This separate format is therefore not examined in this study. Throughout this paper, the term "PC" refers to the combined format of PC.

IAS 31 (1990) also has specific requirements for the supplementary disclosure in the notes to the accounts. It requires that a venturer using EM or the combined format of PC also disclose the aggregate amounts of current assets, non-current assets, current liabilities and non-current liabilities and income and expenses related to its interests in JCEs. These additional disclosures closely resemble those that are otherwise presented on the face of the financial statement if the separate format of PC is adopted. No supplementary disclosure in the notes to the accounts is therefore required for the separate PC format in IAS 31 (1990). In this paper, which limits its scope to EM and the combined format of PC, there is therefore no difference in their supplementary footnote disclosure requirements.

Although EM and PC differ only in their means of aggregation, the resulting accounting amounts presented in financial statements are, however, different except for the bottom-line profit and shareholders' equity, which are the same regardless of whether EM or PC is used. This study focuses on the key accounting amounts of total operating income (or revenue), total operating expenses, other income and expenses (net) and total assets and liabilities, which are intrinsically different under EM and PC.

2.2. Arguments for and against

The arguments for and against the two methods are discussed at great length in a special report "Reporting Interests in Joint Ventures and Similar Arrangements," issued by the U.S. FASB in cooperation with the G4 + 1⁷ in September 1999 (Milburn and Chant, 1999). There are two main issues in the arguments: (1) the control debate, and (2) the need for informativeness about the assets and liabilities of JCEs. The former revolves around the accounting definitions of assets and liabilities and tends to reject PC, while the latter assumes a user's perspective and tends to disapprove of EM. Testing definition-based arguments is difficult because it depends on the validity of the definitions themselves. This paper aims to clarify the second issue, informativeness, by testing empirically whether accounting amounts reported in PC financial statements provide a higher association with share prices than those in EM financial statements. A higher association indicates that the information is more relevant to the investors (as the major group of users of financial statements) and is therefore more informative.⁸

2.2.1. Control debate

The two methods can be compared in terms of their consistency with the definitions of an asset in terms of "control" and a liability as a present obligation. Such comparison tends to disprove PC. This is also the main concern raised in ED 9 (2007): by adding the venturer's share of assets and liabilities line-by-line horizontally to those of the venturer's, "PC can lead to the recognition of assets that are not controlled and liabilities that are not obligations" (IASCF, 2007b Paragraph BC 8). A venturer shares control of the activities of the JCE via "shared decisions" only. ED 9 (2007) stresses that "when a party has an interest in only a share of the outcome generated by the activities of a group of assets and liabilities that it jointly controls, the only asset it controls is its investment in the joint venture" (IASCF, 2007c Paragraph IN8). This is the only asset that is recognized using EM.

⁶ None of the sample companies in this study follows the "separate" format of PC.

⁷ G4 + 1 was an informal group formed in the early 1990 s, initially comprising representatives of the Australian Accounting Standards Board, Canadian Accounting Standards Board, United Kingdom Accounting Standards Board, U.S. FASB and observation by the former International Accounting Standards Committee (IASC, reorganized into the International Accounting Standards Board, IASB, in 2001). It aimed to work cooperatively on new issues in financial reporting and to eliminate some of the accounting differences between countries. Its activities came to a halt in 2001 when the former IASC was restructured into the IASB, which is designed to include an active partnership with national standard setters to achieve global convergence.

⁸ "Joint venture" in ED 9 (2007) is essentially the same as "JCE" in IAS 31 (1990), where the venturer's interest in the assets and liabilities of the JCE are indirect. This paper continues to use "JCE" because it is more descriptive and less generic.

⁹ The "expanded EM" of accounting presents the venturer's proportionate interest in major classes of assets, liabilities, revenues and expenses (e.g., property, plant and equipment, intangibles, current assets, cost of sales, operating expenses, etc.) of a JCE as separate line items within the venturer's financial statements (Dieter, Wyatt, & Reklau, 1978). The "expanded equity method" is equivalent to the "separate" format of proportionate consolidation in IAS 31 (1990).

2.2.2. Informativeness debate

The vertically aggregated consolidation of EM has been criticized as uninformative in conveying the scope of operation and the assets and liabilities of the venturer because it reports its share of the JCE net asset only in one line. EM is sometimes regarded as enabling a venturer to avoid balance sheet recognition of its share of the JCE's individual assets and liabilities, to hide debts and facilitate off-balance-sheet activities (e.g., Milburn and Chant, 1999). Similarly, Penman (2007) and White, Sondhi and Fried (2003) comment that EM income statements lack detail on the venturer's share of JCE revenues and expenses, and effective financial analyses are potentially hindered as a result.

Mathematically, EM tends to improve financial ratios. For example, by including the venturer's share of JCE earnings in the profit for the period while at the same time excluding the sales revenue and asset base used to generate these earnings, EM reports a higher profitability margin and return on assets although return on equity is unaffected. Similarly, netting joint venture assets and liabilities in the "interests in JCE" account results in a lower leverage ratio, and shielding JCE interest expense in the one-line presentation of "share of JCE profit" results in higher interest coverage (Bauman, 2003). This is illustrated in Panel C of Exhibit 1.

Proponents of PC strongly believe that because PC involves less aggregation (line-by-line horizontal aggregation), it provides more useful information (e.g., Feld et al., 2008). Thus, the issue for the proponents of PC is not whether there are faults in the underlying conceptual validity, but whether the resulting accounting information is more useful for its users.

The informativeness debate is clearly an empirical question. This paper aims to provide an answer to this question by testing empirically the value relevance of EM versus PC statements.

3. Related studies

Practitioners and academics have discussed the issues concerning the concepts and procedures underlying the alternative accounting methods for joint ventures since the 1960s. Kocan (1962), Nielsen (1965) and Reklau (1977) suggest that PC should be used if joint ventures are a way to extend the venturers' activities, and the joint venture debts are in reality venturers' obligations. Dieter et al. (1978) recommend their "expanded EM," which is essentially PC in the separate format. Neuhausen (1982) points out that EM may not provide useful information about the venturer's resources and obligations arising from its JCE investment. These viewpoints are largely consistent with the preference for PC in IAS 31 (1990) that "the substance and economic reality are reflected in the consolidated financial statements of the venturer when the venturer recognizes its pro-rata share of interests in the assets, liabilities, income and expenses of the JCE using ... PC" (IASCF, 2010a Paragraph 32).

Bierman (1992) uses finance theory to argue that PC is superior and should be used for all material intercorporate common stock investments, even majority-owned subsidiaries. King and Lembke (1994) also consider the contentious issue of joint venture accounting, recommending EM when the operations of the venturer and the joint venture are dissimilar and PC when they are similar. Davies and Largay (1999) perform a hypothetical analysis but find "no substantive justification for continued use of the EM...due to the method's intrinsically limited informational characteristics" (Davies and Largay, 1999, p. 281). Instead, they recommend PC in two formats, combined or separate (labeled as "expanded EM"), as in IAS 31 (1990), depending on whether the operations of the venturer and the joint venture are similar or dissimilar. In contrast, the earlier mentioned FASB/G4+1 report (Milburn and Chant, 1999) reviews and concludes that EM is the more appropriate method, primarily because of the conceptual fault of the PC alternative. Nobes (2002) criticizes EM for its lack of independent theoretical justification, and warns that the pressure for international harmonization ("harmonization haste") can lead to worldwide use of bad methods as well as good ones, an issue also addressed in Miller and Leo (1997).

Addressing the concern that EM promotes off-balance-sheet activities, Bauman (2003) proxies the magnitude of off-balance-sheet activities concealed under EM by the incremental amount of assets/liabilities under pro forma PC. The finding that market participants place greater weight on off-balance-sheet liabilities than assets for companies that provide explicit guarantees of investee obligations provides some empirical support for PC in the financial statements.

There is also concern about the conceptual validity of using the same EM to account for both investments in JCEs and associates if the option of PC is removed for JCEs. Joint control typically envisages a higher degree of influence than that of a "significant influence" over an associate, although this is not supported by the results of related empirical studies (O'Hanlon and Taylor, 2007; Soonawalla, 2006).

To date, there are few empirical studies directly comparing EM and PC, and those that do focus on the implications for the venturer's bond rating arising from the differential off-balance-sheet nature of the reported interests in the joint venture under the two methods. Using a sample of Canadian firms, Kothavala (2003) finds that accounting ratios such as leverage, return on assets and profit margin produced from EM statements are more relevant for bond ratings than are those from PC statements. As the findings are surprising, Bauman (2007) re-examines the issue using a sample of U.S. manufacturing firms and finds, in contrast, that PC ratios have greater bond rating relevance. He attributes the differing results to the greater sample homogeneity in his study. An earlier study by Stoltzfus and Epps (2005), using a sample of U.S. firms, also finds that selected PCcomputed accounting ratios are more strongly associated with bond risk premiums, but the strong association is only for firms that guarantee the debt of their joint venture investments. Graham et al. (2003) examine the predictability of profits and find that PC provides more informative DuPont ratios than EM for predicting future returns on shareholders' equity (book/accounting based) for a set of Canadian firms. Lourenco and Dias Curto (2010) examine the determinants of the management choice between EM and PC and their results support their prediction that venturers are more likely to change to PC when the majority of their JCEs are "linked" cases; i.e., when each venturer contributes its own critical resources to enter a new business and each controls its own share of the JCE assets and liabilities.

To the best of our knowledge, no study has directly compared the relevance of the two accounting practices for share pricing in the capital market (i.e., value relevance). Following Ball and Brown (1968), the examination of share price behavior is deemed to be an effective way to study investors' behavior as a group, and an accounting number is value relevant to an investor if it is capable of influencing that investor's decision, and has a predicted significant relation with the share price (Amir et al., 1993; Barth et al., 2001; Hu et al., 2014). Brown and Howieson (1998) identify a number of areas for future capital research in an accounting standard setting, one of them being joint venture accounting. This study aims to address this apparent gap in the literature and seeks to understand the value relevance of the two alternative accounting practices for joint ventures.

A review of the studies by Kothavala (2003), Graham et al. (2003), Stoltzfus and Epps (2005) and Bauman (2007) raises a common question about their research design—the use of artificial "pro forma" financial statements and the accounting ratios computed from them. These studies are conducted in either the U.S. or Canada, where only one method is allowed (Canada generally requiring PC and the U.S. requiring EM). To make comparisons between the two alternative methods, accounting numbers and ratios are computed from pro forma statements constructed for the other method "as if" the users have full knowledge about the differences in the two methods and have correctly made all of the necessary adjustments accordingly. Unlike these earlier studies, this study utilizes the uniqueness of the regulatory financial reporting setting in Hong Kong for accounting years beginning on or after 1 January 2005, when a choice between EM and PC was available. The use of differential reporting practices in Hong Kong enables us to examine the two alternative methods in a "real" setting where the "true" share price implications are known and measurable.

4. Research method

Value relevance research typically examines the association between accounting amounts and share price levels or firm value in aggregate (Landsman, 1986; Barth, 1994). Examining changes in share price levels, or returns, is an alternative approach to assessing value relevance (Easton and Harris, 1991; Amir et al., 1993). This study examines the association with both firm value and returns. Consistent results for both associations allow us to be more confident about the robustness of the subsequent inferences, despite the econometric limitations arising from the use of price-based models (Christie, 1987; Kothari and Zimmerman, 1995).

¹⁰ The econometric limitations of the price-based valuation and returns model are largely due to dependencies between the error term of the regression and the independent variables (Christie, 1987). These dependencies may be due to errors in variables, correlated omitted variables or variation in coefficients across observations (Christie, 1987).

Another frequently used model for value relevance research is that based on Ohlson (1995), which conveniently incorporates both the balance sheet and income statement effects in a single equation (Barth et al., 2001). Nevertheless, we do not use Ohlson's (1995) model in this study because its "clean surplus" assumption¹¹ is more restrictive than the accounting standard requirement (Barth 1994) and may not reflect the real situation.

The difference between EM and PC is not confined to just a few specific financial statement items. The difference is wide-ranging and simultaneously affects almost all of the reported items, except the bottom-line profit and shareholders' equity (or net assets). Because it is practically impossible to include all of the affected items, we focus on the key summary statement items that are affected by whether EM or PC is used. For the return model, we include total operating income (or revenue), total operating expenses and other income/expenses (net) reported on the income statements. For the firm value model, we include total assets and total liabilities reported on the balance sheets. By including a dummy variable in both models to indicate whether EM or PC is used during the accounting year, the value relevance of the two methods can be compared.

Control variables are also included in our regressions. To ensure that our results fully capture the effects from the key summary statement items, which are intrinsically different between EM and PC, we ensure that the control variables are measured in such a way that they are the same whether EM or PC is used.

Following Fan and Wong (2002) and Haw et al. (1999), we include typical control variables of firm size, growth opportunity, industry and accounting year. Because the use of EM or PC in the financial statements changes the amounts reported for total assets and total revenue, this study uses market capitalization to measure firm size. For the same reason, growth opportunity is measured by the market-to-book value of equity, which is not affected by the JCE method used.

Because previous studies suggest the possible impact of the operational similarity between a JCE and a venturer (King and Lembke, 1994; Davies and Largay, 1999), we also include control variables indicating whether the venturer provides a guarantee related to the JCE debt (Stoltzfus and Epps, 2005), whether a significant portion of business is conducted through JCEs (Feld et al., 2008), operational similarity, guarantee provision and the size of JCE interest. The size of JCE interest is measured as the venturer's share of JCE net assets as a proportion of the venturer's net assets, both of which are unaffected by the JCE method used.

While previous studies include leverage and return on assets (Fan and Wong, 2002; Kothavala, 2003; Graham et al., 2003; Stoltzfus and Epps, 2005; Bauman, 2007), this study uses the ratio of the venturer's share of the JCE's total liabilities to the venturer's own total liabilities (i.e., without including the venturer's share of the JCE's total liabilities) and return on equity to avoid the difference introduced by the use of either EM or PC in the financial statements.

Mainland China-based companies (red chips¹² and H-shares¹³) are significant in the Hong Kong stock market. The total market capitalization of these companies in December 2008 was HK\$5608 billion, representing 55% of the total market capitalization of all HKSE equities (51% in 2007, 48% in 2006 and 37% in 2005) (HKSE, 2005, 2006, 2007, 2008). These mainland China-based companies have to comply with HKFRS or IFRS in the same way as all other companies listed on the HKSE. However, because investors may have different criteria for these companies (Cheung et al., 2007), a dummy variable is included to distinguish them from the other Hong Kong-based companies.

¹¹ Ohlson's (1995) model expresses firm value in terms of accounting earnings and equity book value. Ohlson's "clean surplus" assumes that all changes in assets/liabilities (except those arising from dividends and capital contributions) must pass through the income statement, i.e., the change in book value is equal to earnings minus dividends (net of capital contributions).

¹² An H-share company is a company incorporated in mainland China but traded on the HKSE. H-share companies' activities are based in mainland China (HKSE 2004). Their financial statements are in Renminbi, the currency of mainland China, and are prepared following the mainland China accounting standards for reporting in mainland China and the IFRS or HKFRS for reporting in Hong Kong. Their shares are quoted in Hong Kong dollars on the HKSE. According to the statistics published by the HKSE, 1261 companies were listed on the HKSE by the end of 2008, of which 150 were H-share companies (HKSE, 2008).

¹³ A red chip company is a company incorporated outside mainland China and listed on the HKSE. Although red chips are not mainland-China-incorporated, they are controlled by mainland China entities (individuals or companies), by way of direct or indirect shareholdings and/or representation on the board of directors (HKSE 2004). If they are part of a mainland China group, their financial statements are to be consolidated and must be in Renminbi and prepared following the mainland China accounting standards. According to HKSE statistics, among its 1261 listed companies, 95 were red-chip companies (HKSE, 2008).

Although Hong Kong companies are free to choose their accounting year-end dates, 31 December is the most popular date, chosen by around 72% of HKSE companies from 2005 to 2008 (as estimated from Datastream). To ensure a simpler research design, this study includes only companies with 31 December year-ends. The sample years in this study are 2005–2008, covering the four-year period when the two alternative options for JCE accounting are available.

Because the choice between EM and PC is an endogenous decision for each company and the factors that determine the EM-PC choice can also influence the dependent variables in the study, we calculate the inverse Mills ratio ¹⁴ (Heckman, 1976) and include it in both the return and firm value regressions to control for the possible selection bias in the estimations.

Following Haw et al. (1999), to mitigate the scale effect, we deflate all of the variables (except the dummy and ratio variables) in the returns association with the market value of common equity on the first day of the fifth month after the beginning of the accounting year, and those in the firm value association with the number of common shares outstanding on the last day of the fourth month after the end of the accounting year. As HKSE listed companies are required to announce their results and publish their annual reports within four months of the accounting year end, an important part of the analyses in this study is based on the 12-month period extending from the first day of the fifth month after the beginning of the accounting year until the last day of the fourth month after the end of the accounting year, which captures the period between the results announcements. As the sample companies in this study all have 31 December year-ends, the between-results announcement period is the 12-month period from May 1 to April 30 of the next year.

4.1. Return association

The return association approach is based on Beaver's (1981) information perspective on financial reporting, which posits that in response to the release of a company's financial statements, investors will analyze the statements for new and unexpected information and revise their beliefs about the company's future performance, causing movements in the company's share prices and resulting in abnormal returns to investors (as first hypothesized by Ball and Brown, 1968). Thus, financial statement information is informative and decision-useful or value-relevant if it results in abnormal returns to investors, Easton and Harris (1991) confine the financial information to reported income or earnings from the income statement and develop a model relating unexpected earnings to abnormal returns. Easton and Harris (1991) measure unexpected earnings by earnings change and earnings level. Earnings change is identified as the dominant measure, while earnings level is included to reduce the measurement error bias in the regression estimates¹⁵ (Brown et al., 1987; Easton et al., 1991). Amir et al. (1993) devise a simple method to measure an abnormal return as the difference between the company's return during the period and the return on the market portfolio, also known as the market-adjusted rate of return. 16 The market-wide return is removed and the abnormal return thus obtained represents a company-specific return, a major indication of abnormal returns. Barth (1994) extends earnings to earnings components and examines the informativeness of a particular component—securities gains and losses. This study focuses on the key summary components of the venturer's earnings that are intrinsically

¹⁴ The inverse Mills ratio, named after John Mills, is a ratio commonly applied in regression analyses to take account of possible selection biases. Heckman (1976) proposed a 2-stage estimation procedure using the inverse Mills' ratio to overcome possible selection biases. In the first stage, a regression to observe a positive outcome of a particular selection is modeled with a probit model. The estimated parameters are then used to calculate the inverse Mills ratio, which is then included as an additional explanatory variable in the second stage OLS estimation.

¹⁵ Easton and Harris (1991) include earnings change (current earnings – previous earnings) as the dominant measure of unexpected earnings, assuming that earnings follow a random walk and previous earnings represent expected earnings. However, the evidence in Beaver and Morse (1978) suggests that earnings/price ratios are mean-reverting and thus the current earnings level may be a useful alternative measure of unexpected earnings. Brown et al. (1987) also demonstrate that multiple proxies for unexpected earnings may reduce the measurement bias in the regression estimates. Easton and Harris (1991) therefore include both earnings change and earnings level as measures of unexpected earnings, although earnings change is the dominant measure.

¹⁶ Brown and Warner (1980) find the market-adjusted rate of return, despite its simplicity, to perform reasonably well under a wide variety of conditions when compared with the more conventional market model used in Ball and Brown (1968) and other studies (e.g., Ball and Kothari, 1991).

different under EM and PC—total operating income (or revenue), total operating expenses and other income and expenses (net) from the income statements.

Following Easton et al. (1991), Amir et al. (1993) and Barth (1994), the regression equations below are used to evaluate the differential value relevance of EM versus PC key income statement components. The main variables of interest are the interaction variables $\Delta INC \times METHOD$, $\Delta EXP \times METHOD$ and $\Delta OINC/EXP \times METHOD$, which are constructed by interacting the EM versus PC variable of METHOD with the dominant measure of unexpected earnings – the earnings change component variables ΔINC , ΔEXP and $\Delta OINC/EXP$.

AR =
$$a_0 + a_1$$
METHOD + a_2 INC + a_3 EXP + a_4 OINC/EXP + a_5 ΔINC + a_6 ΔEXP
+ a_7 ΔOINC/EXP + a_8 ΔINC × METHOD + a_9 ΔEXP × METHOD + a_{10} ΔOINC/EXP
× METHOD + a_{11} SJCELIAB + a_{12} SJCENA + a_{13} SIMILAR + a_{14} GUAR + a_{15} SIZE
+ a_{16} GROW + a_{17} ROE + a_{18} MCHINA + a_{19} IND + a_{20} YR + a_{21} INVMR (1)

The variables are defined as follows.

AR is the buy-and-hold¹⁷ abnormal return (adjusted for dividends and stock splits) for the 12 months beginning on the first day of the fifth month after the beginning of the accounting year, calculated using the market-adjusted return.

METHOD is a dummy variable that indicates whether EM or PC is used for JCE reporting during the accounting year; METHOD is set to one if PC is used during the accounting year, and zero otherwise.

INC is the company's total operating income (or revenue) in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

EXP is the company's total operating expenses in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

OINC/EXP is the company's other income and expenses (net) in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year,

 Δ INC is the difference between INC in the current and previous accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

 Δ EXP is the difference between EXP in the current and previous accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year,

 Δ OINC/EXP is the difference between OINC/EXP in the current and previous accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

SJCELIAB is the venturer's share of the JCE's total liabilities relative to the venturer's own total liabilities at the end of the accounting year.

SJCENA is the venturer's share of the JCE's net assets relative to the venturer's net assets at the end of the accounting year.

¹⁷ While there is disagreement in the literature regarding the method for calculating long-run abnormal returns because of the inherent right-skewed non-normal distribution problem (Lyon et al., 1999), the choice of method is not as important for the measurement of short-run abnormal returns (Fama, 1998; Jakobsen and Voetmann, 2003). Because a horizon of 3–5 years is referred to in the literature as long-run, the 12-month abnormal returns in this study constitute short-run returns. Commonly used abnormal returns in event studies are buy-and-hold abnormal returns (BHAR) (monthly abnormal returns compounded) and cumulative abnormal returns (CAR) (monthly abnormal returns summed and averaged). Lyon et al. (1999) suggest BHAR as suitable for answering the question of whether sample firms earn abnormal returns over a particular period of time, while CAR is preferred where sample firms persistently earn abnormal monthly returns. Fewer data are collected for BHAR because the compounded monthly returns are simply the annual return; BHAR is therefore used in this study.

SIMILAR is a dummy variable to indicate whether the operations of the majority of the JCEs are similar to those of the venturer/venturer-group during the accounting year. SIMILAR is set to one if the operations are similar and zero otherwise.

GUAR is a dummy variable to indicate whether the venturer/venturer group provides a financial guarantee on the JCE debt. GUAR is set to one if a financial guarantee is provided and zero otherwise.

SIZE is the natural logarithm of the company's market capitalization at the end of the accounting year. GROW is the ratio of the company's market-to-book value of equity at the end of the accounting year.

ROE is the ratio of the company's profit for the accounting year after preferred dividends divided by the equity at the beginning of the accounting year.

MCHINA is a dummy variable to indicate whether the company is mainland China-based during the accounting year. MCHINA is set to one if the company is a red chip or H-share company, and zero otherwise.

IND is a set of eight dummy variables IND1, IND2, IND3, IND4, IND5, IND6, IND7 and IND8 to indicate the company's industry in the accounting year, out of the nine industry sectors classified by the HKSE: conglomerates, consumer goods, energy, industrial goods, information technology, materials, properties and construction, services and telecommunications.

YR is a set of three dummy variables, YR1, YR2 and YR3, to indicate the four accounting years ending 31 December 2005, 2006, 2007 and 2008.

INVMR is the inverse Mills ratio computed from a probit regression of the EM-PC choice model, which includes company characteristic variables that may influence the likelihood of a company adopting EM or PC for their JCE interests:

METHOD =
$$x_0 + x_1$$
SJCELIAX + x_2 SJCENA + x_3 SIMILAR + x_4 GUAR + x_5 SIZE + x_6 GROW + x_7 ROE + x_8 MCHINA + x_9 IND

The focus of this study is on the sign and significance of the differential slope coefficients a_8 , a_9 and a_{10} of the interaction variables $\Delta INC \times METHOD$, $\Delta EXP \times METHOD$ and $\Delta OINC/EXP \times METHOD$. As the dummy variable METHOD is coded 1 when PC is used, support for the higher value relevance of PC over EM is provided if the resulting coefficients a_8 and a_{10} (assuming net other income) are both individually significant and positive and a_9 is significantly negative. Conversely, if a_8 and a_{10} (a_9) turn out to be significantly negative (positive), it will provide support for the higher relevance of EM over PC.

Prediction is more straightforward for the separate value relevance of EM and PC. The value relevance of EM (i.e., when METHOD = 0) is assessed by the individual coefficients a_5 , a_6 and a_7 of Δ INC, Δ EXP and Δ OINC/EXP, respectively. The value relevance of PC is assessed by the individual sums of the coefficients a_5 and a_8 , a_6 and a_9 , and a_7 and a_{10} . The signs of a_5 , a_7 , the sum of a_5 and a_8 , and the sum of a_7 and a_{10} are each expected to be positive, as they are the coefficients of Δ INC and Δ OINC/EXP (assuming net other income). The signs of a_6 and the sum of a_6 and a_9 are expected to be negative as they are the coefficients of Δ EXP.

4.2. Firm value association

An alternative value relevance model expresses firm value as a function of the book values of the firm's asset and liability components (Barth, 1994). Following Barth (1994), and with the control variables and inverse Mills ratio added as in the return association, the following regressions evaluate the differential value relevance of the key summary accounting amounts of total assets and total liabilities available from the EM versus PC statements of financial position.

$$MV = b_0 + b_1 METHOD + b_2 ASSET + b_3 LIAB + b_4 ASSET \times METHOD + b_5 LIAB$$

$$\times METHOD + b_6 SJCELIAB + b_7 SJCENA + b_8 SIMILAR + b_9 GUAR + b_{10} SIZE$$

$$+ b_{11} GROW + b_{12} ROE + b_{13} MCHINA + b_{14} IND + b_{15} YR + b_{16} INVMR$$
(2)

where MV is the market value of common equity on the last day of the fourth month after the end of the accounting year, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

ASSET is the company's total assets at the end of the accounting year, as available from the statements of financial position, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

LIAB is the company's total liabilities at the end of the accounting year, as available from the statements of financial position, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

As in regression Eq. (1), the differential value relevance of PC over EM for the statements of financial position is assessed in Eq. (2) by the differential slope coefficients b_4 and b_5 of the interaction variables ASSET × METHOD and LIAB × METHOD. If PC is of higher relevance, b_4 will be significantly positive and b_5 significantly negative. Conversely, if EM is of higher relevance, b_4 will be significantly negative and b_5 significantly positive.

Similar to the coefficients of Δ INC, Δ EXP and Δ OINC/EXP in regression Eq. (1), the coefficients of ASSET are expected to be positive and those of LIAB are expected to be negative, irrespective of whether EM or PC is used. The coefficients of EM are indicated by b_2 and b_3 while those of PC are indicated by the sum of b_2 and b_4 and the sum of b_3 and b_5 .

5. Empirical results

Data for this study are collected from two sources. The financial statement data are hand-collected from the financial statements published on the website of the HKSE http://www.hkex.com.hk. The Morgan Stanley Capital International (MSCI) Equity Hong Kong Index is used as the proxy for equity market return in Hong Kong. The equity market data, including the MSCI Equity Hong Kong Index, are obtained from Datastream.

5.1. Sample characteristics and descriptive statistics

Companies are eligible for selection if, during the 2005 to 2008 sample period, they had interests in JCEs at the year end(s) and (1) they are listed on the HKSE, (2) their accounting year ended on 31 December 2005, 2006, 2007 and 2008, (3) they are not in the finance and utilities industries, (4) their share of JCE total assets is higher than their share of JCE total liabilities, i.e., their share of JCE net assets is a positive amount, (5) they report positive values for their shareholders' equity, and (6) they did not change their accounting year ends, company names or HKSE listing codes during 2004–2008. ¹⁸

After also excluding the cases with missing data, 398 firm-year observations remain. Extreme observations with less than 3% interest in a JCE (i.e., JCESIZE < 3%) are also excluded to mitigate undue influence from these observations, yielding a final sample of 342 firm-year observations.

Table 1 provides the sample make-up according to the categorical characteristics of companies, as a whole and separated into EM and PC venturers. The number of sample companies available increases from 2005 to 2008 (43 to 136), showing that joint ventures were becoming more popular as a form of business arrangement. Throughout the sample years, more companies applied EM than PC (66% of the firm-years applied EM, and 34% PC), largely because the local Hong Kong Statement of Standard Accounting Practice No. 21 "Accounting for Interests in Joint Ventures" (HKSA, 2001), which was in force before 2005, mandated EM. In 2005, even though the convergence with the IFRS opened up PC as an alternative method, the popularity of EM was maintained among the HKSE listers.

Although not reported in Table 1, our sample includes six companies that changed from EM to PC in 2005 and five companies that changed from PC to EM in 2008. Checking against their annual report disclosures revealed that they made the change for convenience only. They were red chip or H-share companies (so-

¹⁸ We exclude 2004 because the earnings change variables ΔINC, Δ EXP and Δ OINC/EXP in the return regression model require data not only from the current year, but also from the previous year.

Table 1 Sample characteristics 342 firm-years.

	Equity met (number of	hod companies)	Proportionate (number of co		Total (number of companies)		
Year							
2005	27	(63%)	16	(37%)	43	(12%)	
2006	40	(61%)	25	(39%)	65	(19%)	
2007	66	(67%)	32	(33%)	98	(29%)	
2008	92	(68%)	44	(32%)	136	(40%)	
	225	(66%)	117	(34%)	342	, ,	
Industry sectors							
Conglomerates	14	(70%)	6	(30%)	20	(6%)	
Consumer goods	38	(64%)	21	(36%)	59	(17%)	
Energy	19	(70%)	8	(30%)	27	(8%)	
Industrial goods	11	(50%)	11	(50%)	22	(6%)	
Information technology	1	(33%)	2	(67%)	3	(1%)	
Materials	9	(60%)	6	(40%)	15	(4%)	
Properties and construction	47	(75%)	16	(25%)	63	(19%)	
Services	86	(66%)	45	(34%)	131	(38%)	
Telecommunications	0	(0%)	2	(100%)	2	(1%)	
	225	(66%)	117	(34%)	342		
Whether mainland-China-based							
Mainland-China-based	128	(67%)	63	(33%)	191	(56%)	
Non-mainland-China-based	97	(64%)	54	(36%)	151	(44%)	
	225	(66%)	117	(34%)	342		
Whether financial guarantee provided to	JCE debt						
Financial guarantee provided	23	(55%)	19	(45%)	42	(12%)	
Financial guarantee not provided	202	(67%)	98	(33%)	300	(88%)	
	225	(66%)	117	(34%)	342		
Operation similarity between venturer an	nd its JCE(s)						
Similar operations	166	(65%)	91	(35%)	257	(75%)	
Dissimilar operations	59	(69%)	26	(31%)	85	(25%)	
	225	(66%)	117	(34%)	342		

called mainland China-based companies in this study) that had chosen to change so that they could use the same JCE method for reporting in both Hong Kong and mainland China. ¹⁹ As reported earlier, our results suggest that the HKSE listers continued to use EM in 2005 despite the availability of PC. However, this might not have been the case for the mainland China-based HKSE listers. As PC was mandated by the mainland China accounting standards in 2005 (MOFPRC, 2001), mainland China-based companies might have chosen to change to PC in 2005 to avoid keeping two sets of JCE accounts. For the same reason, when the mainland China accounting standards first mandated EM in 2008 (MOFPRC, 2006), mainland China-based HKSE listers might have decided to switch to EM. As a result, the 11 mainland China-based companies that changed methods in 2005 and 2008 clearly did so only for convenience. Unreported results confirm that their inclusion or exclusion has no effect on the results of this study.

The switch to EM of the five mainland China-based companies in 2008 contributed to the slightly increased popularity of EM over the sample years (63% in 2005 to 68% in 2008). The slight increase could also result from some companies believing (or being advised by their auditors) that it would be inevitable that the new standard would allow EM only. PC was therefore avoided by new joint ventures.

About 74% of the sample companies were from the services (38%), properties and construction (19%), and consumer goods (17%) industries. Other than the information technology and telecommunications industries, which had very few companies in the sample, and the properties and construction industry, which showed a

¹⁹ Refer to Notes 13 and 14 for the reporting requirements for red chip or H-share companies in mainland China.

higher application of EM (75% EM versus 25% PC), the popularity of EM versus PC was more or less the same across all of the other industries.

The proportion of mainland China-based companies in our sample was higher than among the HKSE listers (191 out of 343 (56%) versus 245 out of 1261 (19%) among the HKSE listers at 31 December 2008, as detailed in Notes 13 and 14). The relative predominance of mainland China-based companies in our sample is consistent with the likely predominance of mainland China JCEs held (as detailed in Note 5) by our sample companies. Most of the sample companies did not provide a financial guarantee for the JCE debt and had similar operations to their JCEs (88% and 75%). Their overall preference for EM over PC was similar to that

Table 2 Descriptive statistics 342 firm-years.

	N	Mean	SD	Max	Min
Control variables in OLS regressions					
When equity method is used					
Share of JCE liabilities over venturer's own total liabilities (%)	225	35.64	55.53	323.64	0.02
Share of JCE net assets over venturer's net assets (%)	225	13.92	13.45	75.76	3.12
Size (market capitalization at year end Dec 31) (HK\$m)	225	14,805	28,808	200,580	42
Growth (market-to-book value of equity)	225	1.44	1.24	6.60	0.17
Return on equity (%)	225	4.74	23.53	47.91	-89.62
When proportionate consolidation is used					
Share of JCE liabilities over venturer's own total liabilities (%)	117	46.58	117.04	719.66	0.00
Share of JCE net assets over venturer's net assets (%)	117	12.55	9.60	44.23	3.12
Size (market capitalization at year end Dec 31) (HK\$m)	117	9878	27,202	121,971	36
Growth (market-to-book value of equity)	117	1.32	0.96	4.25	0.17
Return on equity (%)	117	6.13	16.77	48.82	-44.06
Dependent variables in OLS regressions					
When equity method is used					
Abnormal return (%)	225	7.02	67.70	344.18	-89.15
Market value of common equity at Apr 30 (HK\$m)	225	15,033	25,567	139,968	84
When proportionate consolidation is used					
Abnormal return (%)	117	2.42	34.53	101.18	-96.33
Market value of common equity at Apr 30 (HK\$m)	117	9551	25,531	116,809	130
Key financial statement variables in ols regressions					
When equity method is used					
Total income (HK\$m)	225	34,139	187,749	1,582,449	30
Total expenses (HK\$m)	225	34,018	191,100	1,645,172	813
Other income and expenses (net) (HK\$m)	225	3836	24,068	229,760	-28,624
Total assets (HK\$m)	225	31,603	109,631	863,815	219
Total liabilities (HK\$m)	225	15,692	59,349	464,156	6
When proportionate consolidation is used					
Total income (HK\$m)	117	61,273	227,351	1,032,874	26
Total expenses (HK\$m)	117	60,115	222,463	1,010,491	23
Other income and expenses (net) (HK\$m)	117	4050	18,527	80,929	-14,862
Total assets (HK\$m)	117	44,907	134,006	608,199	381
Total liabilities (HK\$m)	117	24,845	72,067	321,730	33
Other financial statement data					
When the equity method is used					
Profit for the year (HK\$m)	225	3957	20,194	167,038	-28,730
Net assets (HK\$m)	225	15,910	50,755	399,659	113
Share of JCE profit (HK\$m)	225	138	366	2933	-1091
Share of JCE net assets (HK\$m)	225	1195	2035	13,785	15
When proportionate consolidation is used				,	
Profit for the year (HK\$m)	117	5208	23,387	103,311	-18,309
Net assets (HK\$m)	117	20,062	62,568	286,469	206
Share of JCE profit (JK\$m)	117	248	670	2749	-263
Share of JCE net assets (HK\$m)	117	1209	2669	12,294	14

Table 3
Pearson correlations (except industry and year dummy variables and interaction variables) 342 firm-years.

Return assoc	ciation regre	ssion (1)														
	AR	METHOD	INC	EXP	OINC/ EXP	ΔΙΝC	ΔΕΧΡ	ΔΟΙΝC/ EXP	SJCELIAB	SJCENA	SIMILAR	GUAR	SIZE	GROW	ROE	MCHINA
METHOD	-0.08															
INC	-0.14^{***}	0.14**														
EXP	-0.14**	0.13**	0.99***													
OINC/	0.08	0.12**	0.06	0.12**												
EXP																
ΔINC	0.03	0.10^{*}	0.06	0.05	0.12**											
ΔΕΧΡ	0.01	0.08	0.16***	0.15***	0.13**	0.98***										
ΔOINC/	0.16***	0.07	-0.10^{*}	-0.10^*	0.54***	0.09^{*}	0.09^{*}									
EXP																
SJCELIAB	0.09	0.06	-0.16^{***}	-0.16^{***}	-0.01	-0.11**	-0.12^{**}	0.00								
SJCENA	0.05	-0.05	-0.15^{***}	-0.14^{***}	0.07	-0.07	-0.08	0.03	0.35***							
SIMILAR	-0.04	0.04	0.11**	0.09^{*}	-0.09^{*}	-0.08	-0.08	-0.06	0.09	0.08						
GUAR	0.04	0.09	0.05	0.05	0.10^{*}	0.13**	0.12**	0.13**	-0.11**	-0.11^*	-0.11^{**}					
SIZE	0.12**	-0.18^{***}	-0.17	-0.17	0.16***	0.25***	0.20***	0.08	-0.15***	-0.15^{***}	-0.07	0.23***				
GROW	0.09^{*}	-0.05	-0.16	-0.15	0.10^{*}	0.11**	0.08**	0.19***	0.13**	-0.03	-0.03	0.08	0.57***			
ROE	0.30***	0.03	-0.14^{**}	-0.16***	0.36***	0.12**	0.08**	0.63***	-0.01	-0.02	-0.03	0.08	0.24***	0.20***		
MCHINA	0.15***	-0.03	0.12**	0.12**	0.16	0.12**	0.12**	0.12**	-0.04	-0.15^{***}	-0.14^{***}	0.15***	0.38***	0.15***	0.16***	
INVMR	-0.10^{*}	0.89***	0.06	0.06	0.16***	0.15***	0.12***	-0.01	-0.01	0.03	-0.01	-0.05	0.06	0.02	-0.01	0.01
Firm value a	ssociation r	egression (2)														
	MV	METHOD	ASSET	LIAB	SJCELIAB	SJCENA	SIMILAR	GUAR	SIZE	GROW	ROE	MCHINA				
METHOD	-0.24^{***}															
ASSET	0.41***	-0.18^{***}														
LIAB	0.16***	-0.10^{**}	0.85***													
SJCELIAB	-0.06	0.06	-0.16^{***}	-0.18^{***}												
SJCENA	0.00	-0.05	-0.07	-0.08	0.35***											
SIMILAR	-0.10^{*}	0.04	0.07	0.03	0.09	0.08										
GUAR	0.05	0.09	0.19***	0.17***	-0.11**	-0.11^*	-0.11**									
SIZE	0.44***	-0.18^{***}	0.35***	0.32***	-0.15^{***}	-0.15^{***}	-0.07	0.23***								
GROW	0.40***	-0.05	0.11	0.16***	0.13**	-0.03	-0.03	0.08	0.57***							
ROE	0.19***	0.03	0.12**	0.03	-0.01	-0.02	-0.03	0.09	0.24***	0.20***						
MCHINA	0.04	-0.03	0.21***	0.26***	-0.04	-0.15^{***}	-0.14^{***}	0.15***	0.38***	0.15***	0.16***					
INVMR	-0.16^{***}	0.89***	-0.06	0.01	-0.01	0.03	-0.01	-0.05	0.06	0.02	-0.01	0.01				

AR = buy-and-hold abnormal return (adjusted for dividends and stock splits) for the 12 months beginning on the first day of the fifth month after the beginning of the accounting year, calculated using market-adjusted return.

METHOD = dummy variable to indicate whether EM or PC is used for JCE reporting during the accounting year, METHOD is set to one if PC is used during the accounting year, and zero otherwise.

INC = company's total operating income (or revenue) in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

EXP = company's total operating expenses in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

OINC/EXP = company's other income and expenses (net) in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

INC = difference between INC in the accounting year and INC in the prior accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

EXP = difference between EXP in the accounting year and EXP in the prior accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

OINC/EXP = difference between OINC/EXP in the accounting year and OINC/EXP in the previous accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

SJCELIAB = venturer's share of the JCEs' total liabilities relative to the venturer's own total liabilities at the end of the accounting year.

SJCENA = venturer's share of the JCEs' net assets relative to the venturer's net assets at the end of the accounting year.

SIMILAR = dummy variable to indicate whether the operations of the majority of the JCEs are similar to those of the venturer/venturer-group during the accounting year. SIMILAR is set to one if the operations are similar and zero otherwise.

GUAR = dummy variable to indicate whether the venturer/venturer group provides a financial guarantee for the JCE debt. GUAR is set to one if a financial guarantee is provided and zero otherwise.

SIZE = natural logarithm of the company's market capitalization at the end of the accounting year.

GROW = ratio of the company's market-to-book value of equity at the end of the accounting year.

ROE = ratio of the company's profit for the accounting year after preferred dividends divided by the equity at the beginning of the accounting year.

MCHINA = dummy variable to indicate whether the company is mainland China-based during the accounting year. MCHINA is set to one if the company is a red chip or H-share company, and zero otherwise.

INVMR = inverse Mills ratio computed from a probit regression for EM-PC choice model which includes company characteristic variables that may influence the likelihood of a company adopting EM or PC for their JCE interests:

METHOD = $x_0 + x_1$ SJCELIAB + x_2 SJCENA + x_3 SIMILAR + x_4 GUAR + x_5 SIZE + x_6 GROW + x_7 ROE + x_8 MCHINA + x_9 IND.

MV = market value of common equity on the last day of the fourth month after the end of the accounting year, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

ASSET = company's total assets at the end of the accounting year, as available from the statements of financial position, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

LIAB = company's total liabilities at the end of the accounting year, as available from the statements of financial position, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

- * Significant at 10%.
- ** Significant at 5%.
- *** Significant at 1%.

in the entire sample (64–69% EM versus 31–36% PC), except when a financial guarantee was provided for the JCE debt; in these cases, the proportion applying PC was higher (55% EM versus 45% PC), which would be favorable to investors. When a financial guarantee was provided for the JCE debt, investors would probably prefer to have some information about the company's share of the JCE's individual assets and liabilities, and PC would easily make this information available to them.

Table 2 contains the descriptive statistics for the 342 firm-years in our sample, separated into EM and PC companies. On average, the companies that applied EM in our sample had a higher market capitalization but smaller financial statement numbers (market capitalization HK\$14,805 m-15,033 m, total assets/liabilities HK \$31,603 m/15,692 m and total income/expenses HK\$34,139/34,018) than those that applied PC (HK\$9551 m-9878 m, HK\$44,907 m/24,845 m, and HK\$61,273 m/60,115 m, respectively). They also, on average, had slightly higher growth and lower return on equity and offered their investors slightly higher abnormal returns (market-to-book value of equity 1.44, return of equity 4.74% and abnormal return 7.02%) than PC companies (1.32, 6.13%, and 2.42%, respectively). The scale of JCE interests held in our sample was, however, more or less similar on average, irrespective of whether the company applied EM or PC, as indicated by the share of JCE net assets over total net assets of 13.92% for EM versus 12.55% for PC; share of JCE liabilities over venturer's own total liabilities of 35.64% for EM versus 46.58% for PC; dollar amount for share of JCE profit of HK\$138 m for EM versus HK\$248 for PC; and dollar amount for share of JCE net assets of HK\$1195 for EM versus HK\$1209 for PC.

5.2. Correlation analysis

Table 3 reports the zero-order Pearson product moment corrections between the regression variables except the industry and year dummies and the interaction variables.

Among the regression variables, MV, INC, EXP, OINC/EXP, Δ INC, Δ EXP, Δ OINC/EXP, ASSET and LIAB are scaled variables. Their correlations with the other variables, especially the unscaled ones, are difficult to interpret. The discussion in this section therefore focuses more on the unscaled variables. The scaled variables MV, INC, EXP, OINC/EXP, Δ INC, Δ EXP, Δ OINC/EXP, ASSET and LIAB are included in the correlation analysis mainly to detect multicollinearity.

On the whole, the correlations reported in Table 3 are relatively low, although many are significant. The only exceptions are between INC and EXP (r = 0.99), Δ INC and Δ EXP (r = 0.98), ASSET and LIAB (r = 0.85) and METHOD and INVMR (0.89). These high correlations are understandable because they reflect the inclusion of variables that tend to change together or with another common firm characteristic (INC and EXP; Δ INC and Δ EXP; and ASSET and LIAB tend to change together as firm size changes) or are alternative measures of a certain firm characteristic (METHOD measures the actual JCE method used while INVMR is a statistically generated variable measuring the probability of a JCE method being selected based on firm characteristics that may influence the selection). In this study, tolerance and variance inflation factors are also assessed as part of the regression analysis to detect multicollinearity (Gujarati, 2008).

The correlations with AR are largely consistent with those with MV. Both AR and MV are negatively correlated with SIMILAR, suggesting that companies with JCEs with similar operations offer a lower AR and have a lower MV. Both AR and MR are, however, positively correlated with GUAR, SIZE, GROW, ROE and MACHINE, suggesting that companies that provide a guarantee for the JCE debt are larger and have higher growth and ROE, and companies that are mainland China-based offer higher abnormal returns and have a higher market price (measured as the market value of common equity scaled by the number of common shares outstanding). As expected, AR is positively correlated with ΔINC while MR is positively correlated with ASSET (measured as total assets scaled by the number of common shares outstanding).

The correlations of METHOD with the other variables (other than the scaled variables INC, EXP, OINC/EXP, Δ INC, Δ EXP, Δ OINC/EXP, ASSET and LIAB) are consistent with the earlier results from the descriptive statistics. For example, the significant negative correlation with SIZE (measured by natural logarithm of market capitalization) reflects that the EM companies have higher market capitalization.

5.3. Regression analysis

Table 4 reports the results from the return association and the firm value association models. The results from both models provide consistent evidence that EM has higher value relevance than PC for JCE reporting. The key summary income statement and statement of financial position amounts available from EM statements show higher associations with share prices than those from PC statements. This lends support to the elimination of PC.

Between the two models, a higher overall association, indicated by higher R^2 values and lower standard errors of residuals, is found with the firm value model. This is consistent with the typical findings in other studies that include both associations (e.g., Amir et al., 1993; Barth, 1994, Barth and Clinch, 1998; Aboody et al., 1999).

5.3.1. Return association

The results from the return association regression provide strong and consistent evidence that PC is of differentially lower relevance to investors than EM.

The signs and significance of the coefficients a_8 , a_9 and a_{10} of the interaction variables $\Delta INC \times METHOD$ (t = -2.26, p = 0.024), $\Delta EXP \times METHOD$ (t = 2.22, p = 0.028) and $\Delta OINC/EXP \times METHOD$ (t = -1.87, p = 0.062) from estimating Eq. (1) indicate that the key summary amounts from the PC income statement are of significantly lower value relevance than those from the EM income statements.

The lower value relevance of PC statements can also be observed in the change from a significant result for the EM income statements to a non-significant result for the PC income statements when they are assessed separately. The coefficients a_5 , a_6 and a_7 of the respective variables Δ INC, Δ EXP, and Δ OINC/EXP in Eq. (1) represent the value relevance of the EM income statements alone (i.e., when METHOD = 0). They are found to be significant and their signs are as expected (a_5 : t = 4.44, p = 0.000; a_6 : t = -4.24, p = 0.000; and a_7 : t = 1.91, p = 0.057). However, when the PC income statements are assessed separately by running Eq. (1) again but with METHOD set to 0 when PC is used, no significant results are observed (Δ INC: t = -0.14, p = 0.892; Δ EXP: t = 0.20, p = 0.841; and Δ OINC/EXP: t = -0.22, p = 0.829).

5.3.2. Firm value association

The results from the firm value association regression are consistent with those from the return association regression: both are in favor of EM.

The signs and significance of the coefficients b_4 and b_5 of the interaction variables ASSET × METHOD (t = -2.15, p = 0.032) and LIAB × METHOD (t = 1.86, p = 0.064) from estimating Eq. (2) indicate that the key summary amounts from the PC statements of financial position are of significantly lower value relevance than those from the EM statements.

As with the income statements, evidence of the lower value relevance of PC can also be observed from the loss of significance for the PC statements of financial position. From Eq. (2), the coefficients b_2 of ASSET and b_3 of LIAB, which summarize the value relevance of the EM statements of financial position, are significant and their signs are as expected (b_2 : t = 14.10, p = 0.000; b_3 : t = -10.99, p = 0.000). However, no significant results are observed when the PC statements of financial position are assessed separately (ASSET: t = 1.46, p = 0.144; LIAB: t = -1.15, p = 0.25).

5.4. Multicollinearity and heteroscedasticity

Measures of tolerance and the variance inflation factor (VIF) suggest that the independent variables in the regression analysis (i.e., INC and EXP, Δ INC and Δ EXP, ASSET and LIAB, and METHOD and INVMR) are subject to multicollinearity, with VIFs of 577.93, 622.76, 102.59, 99.59, 61.51 and 57.54, respectively, all exceeding the commonly applied rule of thumb of VIF = 10 (Hair et al., 2009). In an attempt to mitigate multicollinearity, we re-run the regressions with (1) INC, EXP and OINC/EXP combined into one variable EARN (representing profit for the year), because INC and EXP (together with OINC/EXP) are not our main variables of interest and they are included in the return model to reduce the measurement error (Brown et al., 1987; Easton et al., 1991); (2) Δ EXP and Δ OINC/EXP combined into one variable Δ EXPnetOINC/EXP rep-

Table 4
Regression results.

-	Return association	n Eq. (1) AR	Firm value associa	ation Eq. (2) MV
	а	t	b	t
Intercept	158.35	1.33	-29.01	-1.58
METHOD	-11.71	-0.27	1.55	0.23
INC	-74.93***	-5.45		
EXP	74.42***	5.38		
OINC/EXP	-27.35***	-4.66		
ΔΙΝC	75.24***	4.40		
ΔΕΧΡ	-74.04^{***}	-4.24		
ΔOINC/EXP	14.07*	1.91		
Δ INC × METHOD	-80.16^{*}	-2.26		
Δ EXP × METHOD	81.57**	2.22		
Δ OINC/EXP × METHOD	-15.99***	-1.87		
ASSET			0.88***	14.10
LIAB			-1.16***	-10.99
$ASSET \times METHOD$			-0.55^{**}	-2.15
$LIAB \times METHOD$			0.77**	1.86
SJCELIAB	0.03	0.73	-0.02^{**}	-2.33
SJCENA	0.11	0.34	0.06	1.22
SIMILAR	3.87	0.54	-3.56^{***}	-2.97
GUAR	-10.02	-0.64	-2.84	-1.13
SIZE	-5.23	-1.01	1.64**	2.04
GROW	4.27	1.10	4.26***	6.90
ROE	1.15***	5.24	0.02	0.75
MCHINA	15.86**	2.19	-4.68***	-4.22
IND dummies	Included but not reported			
YR dummies	Included but not reported			
INVMR	-1.52	-0.06	-1.59	-0.41
F	6.56***		20.73***	
$Adj R^2$	0.33		0.59	
N	342		342	

AR = buy-and-hold abnormal return (adjusted for dividends and stock splits) for the 12 months beginning on the first day of the fifth month after the beginning of the accounting year, calculated using market-adjusted return.

MV = market value of common equity on the last day of the fourth month after the end of the accounting year, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

METHOD = dummy variable to indicate whether EM or PC is used for JCE reporting during the accounting year. METHOD is set to one if PC is used during the accounting year, and zero otherwise.

INC = company's total operating income (or revenue) in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

EXP = company's total operating expenses in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

OINC/EXP = company's other income and expenses (net) in the accounting year, as available from the income statements, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

INC = difference between INC in the accounting year and INC in the prior accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

EXP = difference between EXP in the accounting year and EXP in the prior accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

OINC/EXP = difference between OINC/EXP in the accounting year and OINC/EXP in the prior accounting year, scaled by the market value of common equity on the first day of the fifth month after the beginning of the accounting year.

ASSET = company's total assets at the end of the accounting year, as available from the statements of financial position, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

LIAB = company's total liabilities at the end of the accounting year, as available from the statements of financial position, scaled by the number of common shares outstanding (adjusted for dividends and stock splits) on the last day of the fourth month after the end of the accounting year.

SJCELIAB = venturer's share of the JCEs' total liabilities relative to the venturer's own total liabilities at the end of the accounting year. SJCENA = venturer's share of the JCEs' net assets relative to the venturer's net assets at the end of the accounting year.

SIMILAR = dummy variable to indicate whether the operations of the majority of the JCEs are similar to those of the venturer/venturer-group during the accounting year. SIMILAR is set to one if the operations are similar and zero otherwise.

GUAR = dummy variable to indicate whether the venturer/venturer group provides a financial guarantee for the JCE debt. GUAR is set to one if financial guarantee is provided and zero otherwise.

SIZE = natural logarithm of the company's market capitalization at the end of the accounting year.

GROW = ratio of the company's market-to-book value of equity at the end of the accounting year.

ROE = ratio of the company's profit for the accounting year after preferred dividends divided by the equity at the beginning of the accounting year.

MCHINA = dummy variable to indicate whether the company is mainland China-based during the accounting year. MCHINA is set to one if the company is a red chip or H-share company, and zero otherwise.

IND = set of eight dummy variables IND1, IND2, IND3, IND4, IND5, IND6, IND7 and IND8 to indicate the company's industry in the accounting year, out of the nine industry sectors as classified by the HKSE – conglomerates, consumer goods, energy, industrial goods, information technology, materials, properties and construction, services and telecommunications.

YR = set of three dummy variables, YR1, YR2 and YR3, to indicate four accounting years – year ended 31 December 2005, 2006, 2007 and 2008.

INVMR = inverse Mills ratio computed from a probit regression of the EM-PC choice model, which includes company characteristic variables that may influence the likelihood of a company adopting EM or PC for their JCE interests:

METHOD = $x_0 + x_1$ SJCELIAB + x_2 SJCENA + x_3 SIMILAR + x_4 GUAR + x_5 SIZE + x_6 GROW + x_7 ROE + x_8 MCHINA + x_6 IND

- $+ x_9$ IND. * Significant at 10%.
- ** Significant at 5%.
- *** Significant at 1%.

resenting total operating expenses net of other income and expenses, because $\Delta OINC/EXP$ is not expected to change with firm size as much as ΔEXP and ΔINC do; and (3) LIAB dropped from the firm value regression and INVMR dropped from both the return and firm value regressions because they replicate, to some extent, the variables ASSET and METHOD, respectively, which are already included in the regressions. The new VIFs from the re-runs show that the multicollinearity problem is largely eliminated by these changes. The new VIFs of EARN, ΔINC , $\Delta EXPnetOINC/EXP$, and METHOD for the return model are reduced to 2.22, 11.29, 11.60 and 1.36 respectively; and the new VIFs of ASSET and METHOD for the firm value model are reduced to 1.41 and 2.08, respectively. The new results with the changes made to the variables are similar and continue to show that EM is of higher relevance, although the significance is generally weaker (for $\Delta INC \times METHOD$: t = -1.65, p = 0.10; $\Delta EXPnetOINC/EXP \times METHOD$: t = 0.18, p = 0.08; ASSET $\times METHOD$: t = -2.16, t = 0.03). We therefore make no adjustment for multicollinearity because the inclusion of INC and EXP, ΔINC and ΔEXP , ASSET and LIAB, and METHOD and INVMR, and the consistent results we obtain from them, increase the credibility of our results.

As heteroscedasticity often occurs in cross-sectional and time-series datasets, we check for it using White's procedure (White, 1980). Unreported *t*-scores based on White's (1980) heteroscedasticity-consistent standard errors are slightly lower for most of the independent variables than those reported in Table 4, which are based on OLS *t*-scores. All parameters remain significant, suggesting that heteroscedasticity is not a problem in this data set.

5.5. Sensitivity analysis

Morgan Stanley Capital International Equity Hong Kong is used as the proxy for equity market return in Hong Kong in the calculation of market-adjusted rates of return for the return association regression. We conduct a sensitivity analysis using the Hong Kong Hang Seng Composite Index as the proxy. Except for the adjusted R², which is lowered by 1.47%, the results for the return association are almost the same as those obtained with the Morgan Stanley Capital International Equity Hong Kong proxy. The Hong Kong Hang Seng Composite Index covers 90% of the market capitalization of the shares listed on the Main Board of the HKSE and there are currently 200 constituent shares in this index.

We also examine whether our results are different if we drop the criterion for including only those observations with JCE net assets of $\geq 3\%$. We rerun both the return and firm value association regressions, first using larger samples of 380 and 366 observations with lower cutoffs of $\geq 1\%$ and $\geq 2\%$, and then using smaller samples of 272 and 227 observations with higher cutoffs of $\geq 5\%$ and $\geq 7\%$. The results do not vary signifi-

Table 5
Natural experiment results.

	Return associa	ation Eq. (1) AR	Firm value associ	ation Eq. (2) MV
	а	t	b	t
Intercept	0.36^{*}	1.91	-86.77***	-23.25
D2008	-0.15***	11.72	0.67	-1.39
INC	-21.07^{***}	3.12		
EXP	83.74***	10.61		
OINC/EXP	-26.31^{***}	-6.62		
ΔΙΝC	64.86***	7.15		
ΔΕΧΡ	-92.30^{***}	8.79		
ΔOINC/EXP	8.89***	9.83		
Δ INC × D2008	-61.14^{***}	-2.79		
$\Delta EXP \times D2008$	93.69***	7.43		
Δ OINC/EXP \times D2008	-6.66^{***}	-7.26		
ASSET			0.31***	4.33
LIAB			-0.92^{***}	-8.19
$ASSET \times D2008$			-0.23***	-7.28
$LIAB \times D2008$			0.78***	6.09
SIZE	6.51	-1.01	1.21**	24.76
GROW	-3.61***	1.10	4.80***	15.31
ROE	1.05***	5.24	0.28***	10.82
INDUSTRY dummies	Included but not reported			
$Adj R^2$	0.10		0.79	
N	97,449		97,449	

Note: D2008 = A dummy variable equals one if the fiscal year is before 2008, and zero otherwise.

cantly, and continue to suggest that EM statements are of significantly higher relevance. The only exception to this finding is that when the cutoff is lowered to $\geq 1\%$, the coefficients of the interaction variables in the firm association regression are no longer significant (b_4 of ASSET × METHOD: t = -1.32, p = 0.19; and b_5 of LIAB × METHOD: t = 0.83, p = 0.41). In practice, however, it is uncommon for companies to have such small JCEs: only 18 companies (i.e., 4.52%) of the 398 companies in our sample have JCEs with relative net assets $\geq 1\%$.

5.6. Additional test

To address the potential endogeneity concern, such as omitted variable issues, we use the replacement of EM over PC in mainland China as a research setting to conduct a difference-in-differences analysis. In mainland China, companies used PC following the "Interim Standards on Consolidated Financial Statements" before 2008. In 2008, because of the convergence to International Accounting Standards, PC was eliminated and EM was mandated from then on. This provides an appealing research setting for conducting a quasinatural experiment to test whether PC or EM leads to more informative accounting information. We compare the value relevance before and after the replacement of PC by EM. Specifically, to test our research questions, we regress the following difference-in-difference models:

$$AR = a_0 + a_1D2008 + a_2INC + a_3EXP + a_4OINC/EXP + a_5\Delta INC + a_6\Delta EXP + a_7\Delta OINC/EXP$$

$$+ a_8\Delta INC \times D2008 + a_9\Delta EXP \times D2008 + a_{10}\Delta OINC/EXP \times D2008 + Controls$$
(3)

$$MV = b_0 + b_1D2008 + b_2ASSET + b_3LIAB + b_4ASSET \times D2008 + b_5LIAB \times D2008 + Controls$$
 (4)

D2008 is a dummy variable that equals one if fiscal year is before 2008, and zero otherwise. As Table 5 shows, the coefficients on $\Delta INC \times D2008$, $\Delta INC \times D2008$ and $\Delta INC \times D2008$ are significantly negative, positive and negative, respectively. This is consistent with the findings in Table 4. Moreover, the coefficients on ASSET \times D2008 and ASSET \times D2008 are significantly negative and positive, respectively, again consistent with those reported in Table 4. Overall, the results reported in Table 5 are generally consistent with those reported in Table 4, which strengthens our research findings in Table 4.

6. Conclusions

Part of the debate concerning the proposed removal of PC from IAS 31 (1990) by ED 9 (2007) is based on the view that PC offers a more detailed understanding when a portion of the operations, assets and liabilities of a JCE is horizontally aggregated with the operations, assets and liabilities of the venturer. The results of this study, however, show no evidence that investors benefit from PC financial statements. On the contrary, the higher value relevance of the EM statements found in this study suggests that the vertical aggregation of information into one number does not result in a loss of information to investors. The results show higher returns and firm value associations with the key summary accounting amounts when the statements are prepared using EM. As suggested by Milburn and Chant, 1999, it could be that PC's horizontal aggregation confuses rather than helps investors to understand the venturer's operations. The results therefore support the preference for EM. However, the empirical results should be interpreted carefully because the definitions of "joint venture" under IAS 31 and ED 9 are not exactly the same. Under IAS 31, the definition is based more on the legal form of the joint arrangement, whereas under ED 9 it is mainly driven by the economic substance of the joint arrangement.

The results of this study also have implications for companies around the world that prepare their financial statements using IFRS.

Acknowledgements

We thank Xijia Su (the editor) and Wenfeng Wang (the discussant) for their helpful comments. Feida (Frank) Zhang acknowledges research funding from the National Natural Science Foundation of China (project number: 71790603, 71302036, 71332004)

Appendix A

Exhibit 1. Illustration of alternative accounting methods for JCE reporting

Illustrative financial statements, disclosures and selected financial ratios

					Panel B: Proportionate consolidation		
(Assuming venture company/group has 40% inte		Panel B1: Combined format	Panel B2: Separate format				
	Venture	JCEs					
	company/group						
	\$	\$	\$	\$	\$		
Income	1000	250	1000	1100	1000		
Share of income of JCE					100		
Interest expense	(500)	(70)	(500)	(528)	(500)		
Share of interest expense of JCE					(28)		
Other income/expense (net)	(200)	(80)	(200)	(232)	(200)		
Share of other income/expense					(32)		
(net) of JCE							
			300				
Share of profit of JCE			40				
Profit for the year	\$ 300	\$ 100	\$ 340	\$ 340	\$ 340		

Plant Share of plant of JCE	\$ 2200	\$ 1900	\$ 2200	\$ 2960	\$ 2200 760
Interest in JCE Total non-current assets Cash Share of cash of JCE	600 \$ 2800 450	\$ 1900 250	\$ 2840 450	\$ 2960 550	\$ 2960 450 100
Inventory Share of inventory of JCE Total current assets	650 \$ 1100	450 \$ 700	750 \$ 1100	\$30 \$ 1380	650 180 \$ 1380
Total assets	\$ 3900	\$ 2600	\$ 3940	\$ 4340	\$ 4340
Share capital Retained earnings Total equity	1100 900 \$ 2000	1500 100 \$ 1600	1100 940 \$ 2040	1100 940 2040	1100 940 2040
Bonds payable Share of bonds payable of JCE	1200	600	1200	1440	1200 240
Total non-current liabilities Accounts payable Share of accounts payable of JCE	1200 700	600 400	1200 700	1440 860	1440 700 160
Total current liabilities Total liabilities Total equity and liabilities	700 \$ 1900 \$ 3900	400 \$ 1000 \$ 2600	700 \$ 1900 \$ 3940	860 \$ 2300 \$ 4340	860 \$ 2300 \$ 4340
Supplementary disclosure in notes to accounts Share of non-current assets of JCE	Ψ 2700	4 2 000	\$ 760	\$ 760	\$
Share of non-current liabilities of JCE Share of non-current liabilities of JCE			280 240	280 240	_ _ _
Share of current liabilities of JCE Share of net assets of JCE (optional)			160 640	160 640	_
Share of sales of JCE Share of expenses of JCE Share of results of JCE (optional)			100 60 40	100 60 40	_
Panel C: Selected financial ratios Profit margin			34%	31%	31%
Interest coverage Return on assets			1.7 9%	1.6 8%	1.6 8%
Return on equity Leverage (debt over equity)			17% 0.9	17 % 1.1	17% 1.1

References

Aboody, D., Barth, M.E., Kasznik, R., 1999. Revaluations of fixed assets and future firm performance. J. Account. Res. 36, 161–191. Accounting Principles Board (APB, which in 1973 was replaced by the Financial Accounting Standards Board, FASB), 1971. APB Opinion 18 The equity method of accounting for investments in common stock. APB Opinion, New York, AICPA.

Amir, E., Harris, T.S., Venuti, E.K., 1993. A comparison of the value-relevance of US versus non-US GAAP accounting measures using Form 20-F reconciliations. J. Account. Res. 31, 230–264.

Ball, R., Brown, P., 1968. An empirical evaluation of accounting income numbers. J. Account. Res. 6, 159–178. Ball, R., Kothari, S.P., 1991. Security returns around earnings announcements. Account. Rev. 66 (4), 718–738.

Barth, M.E., 1994. Fair value accounting: Evidence from investment securities and the market valuation of banks. Account. Rev. 69 (1), 1–25

Barth, M.E., Beaver, W.H., Landsman, W.R., 2001. The relevance of the value relevance literature for financial accounting standard setting: Another view. J. Account. Econ. 31, 77–104.

Barth, M.E., Beaver, W.H., Clinch, G., 1998. Revalued financial, tangible, and intangible assets: Associations with share prices and non-market-based value estimates. J. Account. Res. 36, 199–233.

Bauman, M.P., 2007. Proportionate consolidation versus the equity method: Additional evidence on the association with bond ratings. Int. Rev. Financial Anal. 16, 496–507.

Bauman, M.P., 2003. The impact and valuation of off-balance-sheet activities concealed by equity method accounting. Account. Horizons 17 (4), 303–314.

Beaver, W.H., 1981. Financial Reporting: An Accounting Revolution. Prentice Hall, Englewood Cliffs, N.J..

Beaver, W.H., Morse, D., 1978. What determines price-earnings ratios?. Financial Anal. J. 34 (4) 65-76.

Bierman Jr., H., 1992. Proportionate consolidation and financial analysis. Account. Horizons 6 (4), 5-17.

Brown, L., Griffin, P., Hagerman, R., Zmijewski, M., 1987. An evaluation of alternative proxies for the market's assessment of unexpected earnings. J. Account. Econ. 9 (2), 159–193.

Brown, P., Howieson, B., 1998. Capital market research and accounting standard setting. Account. Finance 38, 5-28.

Brown, S., Warner, J.B., 1980. Measuring security price performance. J. Financ. Econ. 8, 205-258.

Cheung, Y.L., Connelly, J.T., Limpaphayom, P., Zhou, L., 2007. Do investors really value corporate governance? Evidence from Hong Kong market. J. Int. Financial Manage. Account. 18 (2), 86–122.

Christie, A., 1987. On cross-sectional analysis in accounting research. J. Account. Econ. 9, 231-258.

Davies, M.L., Largay, J.A., 1999. Financial reporting of significant-influence equity investments: Analysis and managerial issues. J. Managerial Issues 11 (3), 280–298.

Deloitte, 2008. IASB agenda project: Joint ventures. IAS PLUS, available at http://www.iasplus.com/agenda/jv.htm.

Dieter, R., Wyatt, A.R., Reklau, D.L., 1978. The expanded equity method – An alternative in accounting for investments in joint ventures. J. Accountancy 145 (6), 89–94.

Easton, P., Harris, T., 1991. Earnings as an explanatory variable for returns. J. Account. Res. 29 (1), 19-36.

Fama, E.F., 1998. Market efficiency, long-term returns and behavioral finance. J. Financ. Econ. 49, 283-306.

Fan, J.H., Wong, T.J., 2002. Corporate ownership structure and the informativeness of accounting earnings in East Asia. J. Account. Econ. 33, 401–425.

Feld, K.P., Breker, N., 2008. Comment letter from The Institute Accountants in Germany (IDW). Comment on IASB Exposure Draft ED 9 Joint Arrangements, available at http://www.idw.de/idw/download/Comment Letter ED9.pdf?id=425608&property=Datei.

Graham, R.G., King, R.D., Morrill, C.K.J., 2003. Decision usefulness of alternative joint venture reporting methods. Account. Horizons 17 (2), 123–137.

Gujarati, D.N., 2008. Basic Econometrics, fifth ed. McGrawHill Irwin.

Hair, J.F., Anderson, R.E., Tatham, R.L., Black, W.C., 2009. Multivariate Data Analysis with Readings, sixth ed. Prentice Hall.

Haw, I.M., Oi, D.A., Wu, W., 1999. Value relevance of earnings in an emerging capital market: The case of A-shares in China. Pacific Econ. Rev. 4 (3), 337–347.

Heckman, J.J., 1976. Sample selection as a specification error. Econometrica 47 (1), 153-161.

Hong Kong Institute of Certified Public Accountants (HKICPA), 2004. HKAS 31 Interests in joint ventures. Hong Kong Financial Reporting Standards, available at http://www.hkicpa.org.hk/hksaebk/HKSA_Members_Handbook_Master/volumeII/hkas31.pdf.

Hong Kong Stock Exchange (HKSE), 2005. HKSE Fact Book 2005. HKSE, Hong Kong.

Hong Kong Stock Exchange (HKSE), 2006. HKSE Fact Book 2006. HKSE, Hong Kong.

Hong Kong Stock Exchange (HKSE), 2007. HKSE Fact Book 2007. HKSE, Hong Kong.

Hong Kong Stock Exchange (HKSE), 2008. HKSE Fact Book 2008. HKSE, Hong Kong.

Hong Kong Society of Accountants (HKSA, which in 2004 was renamed as Hong Kong Institute of Certified Public Accountants, HKICPA), 2001. SSAP 21 Accounting for interests in joint ventures. Statement of Standard Accounting Practice, available at http://www.hkicpa.org.hk/professionaltechnical/accounting/rm/withdrawn/ssap21.pdf.

Hu, J., Li, A.Y., Zhang, F.F., 2014. Does accounting conservatism improve the corporate information environment?. J. Int. Account. Auditing Taxation 23 (1), 32–43.

International Accounting Standards Board (IASB), 2011. IASB Work Plan – Projected Timetable as at 1 February 2011, available at http://www.iasb.org/Current+Projects/IASB+Projects/IASB+Work+Plan.htm.

International Accounting Standards Committee (IASC, which in 2001 was renamed as International Accounting Standards Board, IASB), 1990. IAS 31 Financial reporting of interests in joint ventures. International Accounting Standards, London, IASC.

International Accounting Standards Committee Foundation (IASCF, which in 2010 was renamed as International Financial Reporting Standards Foundation, IFRSF), 2007. ED 9 Joint arrangements, IASB Exposure Draft, London, IASCF.

International Accounting Standards Committee Foundation (IASCF, which in 2010 was renamed as International Financial Reporting Standards Foundation, IFRSF), 2007. Basis for conclusion on ED 9 Joint arrangements. IASB Exposure Draft, London, IASCF.

International Accounting Standards Committee Foundation (IASCF, which in 2010 was renamed as International Financial Reporting Standards Foundation, IFRSF), 2007. Introduction on ED 9 Joint arrangements. IASB Exposure Draft, London, IASCF.

International Accounting Standards Committee Foundation (IASCF, which in 2010 was renamed as International Financial Reporting Standards Foundation, IFRSF), 2010. IAS 31 Interests in joint ventures. International Financial Reporting Standards 2010 Bound Volume, London, IASCF.

International Accounting Standards Committee Foundation (IASCF, which in 2010 was renamed as International Financial Reporting Standards Foundation, IFRSF), 2010. Preface to International Financial Reporting Standards. International Financial Reporting Standards 2010 Bound Volume, London, IASCF.

International Accounting Standards Committee Foundation (IASCF, which in 2010 was renamed as International Financial Reporting Standards Foundation, IFRSF), 2010. IAS 28 Investments in associates. International Financial Reporting Standards 2010 Bound Volume, London, IASCF.

International Financial Reporting Standards Foundation (IFRSF), 2010. Conceptual Framework for Financial Reporting. IASCF, London.

Jakobsen, J., Voetmann, T., 2003. Post-acquisition performance in the short and long run: Evidence from the Copenhagen stock exchange 1993–1997. Eur. J. Finance 9 (4), 323–342.

Jiang, C., 2006. A re-examination of technology transfer in Sino-foreign joint ventures after China's WTO accession. J. Am. Acad. Business, Cambridge 9 (2), 133–138.

King, T.E., Lembke, V.C., 1994. An examination of financial reporting alternatives for associated enterprises. Adv. Account. 11, 1–30. Knorr, L., 2008. Comment letter from Accounting Standards Committee of Germany. Comment on IASB Exposure Draft ED 9 Joint Arrangements, available at http://www.standsetter.de/drsc/docs/press_releases/080107_cl_ED9.pdf.

Kocan, P., 1962. Reporting the operations of jointly owned companies. J. Account. 113 (2), 54-59.

Koh, J., Venkatraman, N., 1991. Joint venture formations and stock market reaction: An assessment in the information technology sector. Acad. Manage. J. 34 (4), 869–892.

Kothari, S.P., Zimmerman, J.L., 1995. Price and return models. J. Account. Econ. 20, 155-192.

Kothavala, K., 2003. Proportionate consolidation versus the equity method: A risk measurement perspective on reporting interests in joint ventures. J. Account. Public Policy 22, 517–538.

Landsman, W., 1986. An empirical investigation of pension fund property rights. Account. Rev. 61 (4), 662-691.

Lourenco, I.C., Dias Curto, J., 2010. Determinants of the accounting choice between alternative reporting methods for interests in jointly controlled entities. Eur. Account. Rev. 19 (4), 739–773.

Lyon, J.D., Barber, B.M., Tsai, C.L., 1999. Improved methods for tests of long-run abnormal returns. J. Finance 54 (1), 165-201.

Maes, J.P., 2008. Comment letter from Belgian Accounting Standards Board. Comment on IASB Exposure draft ED 9 Joint arrangements, available at http://www.cnc-cbn.be /NL/News/2008/NL_New15-012008%20BASB%20Comment%20Letter%20on% 20ED%20 9%20Joint %20 Arrangements%20IASB.pdf.

Milburn, J.A., Chant, P.D., 1999. Reporting interests in joint ventures and similar arrangements. Financial Accounting Series No. 201-E Special Report, Norwalk, FASB/G4+1.

Miller, M.C., Leo, K., 1997. The downside of harmonization haste: The equity accounting experience. Aust. Account. Rev. 7 (2), 2–15. Ministry of Finance of the People's Republic of China (MOFPRC), 2001. Investments. Accounting Standards for Business Enterprises, available at http://www.casc.gov.cn/kjfg/ 200611/t20061115_489668.htm.

Ministry of Finance of the People's Republic of China (MOFPRC), 2006. ASBE 2 Long-term equity investments. Accounting Standards for Business Enterprises, available at http://www.casc.gov.cn/kjfg/200607/t20060703 337130.htm.

Neuhausen, B.S., 1982. Consolidation and the equity method - time for an overhaul. J. Account. 153 (2), 54-66.

Nielsen, C.C., 1965. Reporting joint-venture corporations. Account. Rev. 40 (4), 795-804.

Nippa, M., Beechler, S., Klossek, A., 2007. Success factors for managing international joint ventures: A review and an integrative framework. Manage. Organization Rev. 3 (2), 277–310.

Nobes, C., 2002. International development of the equity method. Abacus 38 (1), 16-45.

O'Hanlon, J., Taylor, P., 2007. The value relevance of disclosures of liabilities of equity-accounted investees: UK evidence. Account. Business Res. 37 (4), 267–284.

Ohlson, J.A., 1995. Earnings, book values and dividends in equity valuation. Contemp. Account. Res. 11 (2), 661-687.

Penman, S.H., 2007. Financial Statement Analysis and Security Valuation, third ed. New York, McGraw-Hill/Irwin.

Reklau, D.L., 1977. Accounting for investments in joint ventures - A re-examination. J. Account. 144 (3), 96-103.

Soonawalla, K., 2006. Accounting for joint ventures and associates in Canada, UK and US: Do US rules hide information?. J. Business Finance Account. 33 (3 & 4) 395–417.

Stoltzfus, R.L., Epps, R.W., 2005. An empirical study of the value-relevance of using proportionate consolidation accounting for investments in joint ventures. Accounting Forum 29, 169–190.

White, G.I., Sondhi, A.C., Fried, D., 2003. The Analysis and Use of Financial Statement, third ed. John Wiley and Sons, New York. White, H., 1980. A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity Econometrica. J. Economet. Soc. 48 (4), 817–838.